Author’s response to reviews

Title: PrEvention of posttraumatic contractuRes with Ketotifen 2 (PERK 2) – Protocol for a multicenter randomized clinical trial.

Authors:
Ayoola Ademola (ayoola.ademola@ucalgary.ca)
Kevin Hildebrand (hildebrk@ucalgary.ca)
Prism Schneider (prism.schneider@gmail.com)
Nicholas Mohtadi (mohtadi@ucalgary.ca)
Neil White (nwhite@ucalgary.ca)
Michael Bosse (mbosse@carolina.rr.com)
Alexandra Garven (Alexandra.Garven@albertahealthservices.ca)
Richard Walker (rewalker@ucalgary.ca)
Tolulope Sajobi (ttsajobi@ucalgary.ca)

Version: 2 Date: 05 Feb 2020

Author’s response to reviews:

Dear Assistant Editor,

Below are the responses to each of the comments

1. Title page

Please move your corresponding author information to your title page.

Response: I have moved information about the corresponding author to the title page

2. Trial registration

Please also include the date of trial registration in the Abstract section. If it was not registered prospectively (before enrollment of the first participant), you should include the words 'retrospectively registered'.
Response:

The date of registration of protocol with ClinicalTrials.gov has been stated on the Abstract section.

3. Supplementary information

Please remove the knowledge translation and confidentiality sections from your declarations section as these are not required.

Also, please remove the 'Roles, Responsibility and funding' section, informed consent form and SPIRIT checklist as these are no longer required.

Response:

I have removed the following sections from the manuscript as requested

a) Knowledge translation
b) Confidentiality
c) Roles, Responsibility Funding
d) Informed consent form
e) SPIRIT checklist

Thanks